Author's response to reviews

Title: No evidence for association between tau gene haplotypic variants and susceptibility to Creutzfeldt-Jakob disease

Authors:

Pascual Sanchez-Juan (psanchez@humv.es)
Matthew T Bishop (m.bishop@ed.ac.uk)
Alison Green (alison.green@ed.ac.uk)
Claudia Giannattasio (claudia.giannattasio@iss.it)
Alejandro Arias-Vasquez (a.ariasvasquez@erasmusmc.nl)
Anna Poleggi (anna.poleggi@iss.it)
Richard SG Knight (r.knight@ed.ac.uk)
Cornelia M van Duijn (c.vanduijn@erasmusmc.nl)

Version: 4 Date: 30 November 2007

Author's response to reviews:

MS: 8718001881499065
No evidence for association between tau gene haplotypic variants and susceptibility
to Creutzfeldt-Jakob disease

Dear Prof. Morgenstern,

Thank you very much for accepting our manuscript for publication. We have prepared a revised version following the suggestions made by the Editorial Production Team. Their points have been handled as follows:

1) The list of authors in the manuscript is written as they are in the submission system.
2) Qualifications have been removed from the manuscript and submissions system
3) All text is now black
4) We have changed author(s) to authors
5) We have changed Author¿s contributions to Authors' contributions
6) We have changed 'Foot notes' to Figure Legends
7) Titles have been removed from the figures. The legend and title have been placed after the reference list.
8) We have now cropped the two figures as closely as possible around the image.
9) The website addresses have been removed from the text, and included as references.
Hoping this will meet your approval,

Yours sincerely,

Pascual Sánchez-Juan